Author's response to reviews

Title: A method for encoding clinical datasets with SNOMED CT

Authors:

Dennis H Lee (dlhk@uvic.ca)
Francis Y Lau (fylau@uvic.ca)
Hue Quan (Hue.Quan@albertahealthservices.ca)

Version: 7 Date: 6 September 2010

Author's response to reviews: see over
September 9, 2010

Editor-in-Chief
BMC Medical Informatics and Decision Making

Dear Editor,

We have made the changes outlined in the email from the BMC Series Editorial Production Team. This includes synchronising the abstract and using the authors’ initials in the authors’ contributions section.

Thank you.

Sincerely,

Dennis Lee